' 1999 Franz KurfessVisual Design Guidelines 1Course OverviewuIntroductionuUnderstanding Users andTheir TasksuPrinciples and GuidelinesuInteracting with DevicesuInteraction StylesuUI Design ElementsuVisual Design GuidelinesuUI Development ToolsuIterative Design andUsability TestinguProject Presentations andSelected TopicsuCase StudiesuRecent Developments inHCIDuConclusions ' 1999 Franz KurfessVisual Design Guidelines 2Chapter OverviewVisual Design GuidelinesuMotivationuObjectivesuScreen Layout and DesignuGeneral LayoutuTextuNumbersuCoding TechniquesuColoruImplications for DesignuWindowsuWeb PagesuImportant Concepts andTermsuChapter Summary ' 1999 Franz KurfessVisual Design Guidelines 3Slides FeedbackuI would like to get some feedback from you about thevisual design of the PowerPoint Slides I�m using. Pleasewrite down your impressions (positive or negative) on thefollowing aspects:vGeneral layout-Usage of screen space, arrangements of items, connections betweenrelated slides, ...vText-Font size, type, use of attributes like bold, italicsvColors-Background color, text colors, overall color schemevCoding techniques-Getting attention through blinking, highlighting, underlining, colors, ... ' 1999 Franz KurfessVisual Design Guidelines 4Important UI Design ElementsuWhich of the UI Design elements discussed in thatsection are especially relevant for screen layout anddesign?uuu ' 1999 Franz KurfessVisual Design Guidelines 5MotivationuMost of the interaction between user and computeris guided by the visual presentation of the output onthe screen, and it is very important to present theinformation in a suitable and convenient way.uThe screen layout strongly influences the buildup ofa mental model in the user.uGood screen layout can reduce the effort required bythe user to solve a task, e.g. by using recognitionover recall, by providing clues to legal or expectedinputs, or by signaling the importance of output itemsthrough color or other attributes. ' 1999 Franz KurfessVisual Design Guidelines 6Objectivesuto know about design guidelines for screen layoutand designuto be able to apply these guidelines to the screendesign for specific tasksuto effectively use design techniques for text,numbers, graphics, and other visual elementsuto be able to translate task or user requirements intoappropriate and effective screen layoutsuto coordinate the visual output with other output andinput modes

' 1999 Franz KurfessVisual Design Guidelines 7Evaluation Criteriaurecognition of relevant visual design methodsurecall of the most important visual design methodsuevaluation of screen layout for specific tasksudesign of individual screens for specific functions,and of groups of screens for complete tasks orapplications ' 1999 Franz KurfessVisual Design Guidelines 8Visual Design ' 1999 Franz KurfessVisual Design Guidelines 9Screen Layout and Designuprinciples of screen design apply equally to any kindof interfaceumenus, fill-in forms, displays of database records, GUIs,Web pages, etc.udesigners must work within the constraints imposedby the hardware and software of a systemuscreen sizevnumber of pixels, pixel resolution, screen orientation (horizontal orvertical)udisplay capabilitiesvhighlighting, reverse video, bit-mapped graphics, color, sound[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 10Screen Layout Categoriesufive main categoriesugeneral layoututextunumbersucoding techniquesucolor[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 11General Layout Principlesuscope of informationuarrangement of itemsuformatting standardsufonts and styles[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 12Scope of informationuinclude only information essential to decision makinguavoid putting too much information on the screenunever put more on the screen that absolutely necessaryube brief and concise, but be careful to match the level ofdetail to the user�s knowledge and experienceve.g. displaying on-screen instructions vs. type of user (expert vs.novice)[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 13Scope of Information (cont.)uinclude all information essential to decision makingueach screen should stand alonevcontaining one idea or one taskuusers should never have tov refer to another screen to accomplish a task on the current screenvremember information on one screen to complete anotheruerror messages should be complete, and not requirereference to manuals to decipher or understandvpoor: �file name too long�vbetter: �file name should not exceed 8 characters� ' 1999 Franz KurfessVisual Design Guidelines 14Arrangement of Itemsustart in the upper-left corner of the displayueye-tracking studies show that the eye tends to go to theupper-left corner of a displayuEnglish language is written left to rightvmany other languages do notveye also tends to move in a clock-wise manneruthe human brain tends to seek order and patterns in adisplay, and follow them if they are apparentvGestalt principles-proximity-similarity-closure-continuity-symmetry[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 15Arrangement of Items (cont.)ugroup items logicallyukeep group size manageablevroughly 12-14 characters wide by 6-7 lines high yieldsapproximately 5 deg. of visual angleuminimize the number of groupsvnumber of groups affects search time, even when group size issmalluprovide symmetry and balanceuthrough the use of blank spacesvbut don�t imbed blank spaces to stretch out a word-e.g. use �close,� not � c l o s e �ucenter words if appropriate[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 16Formatting Standardsudesign formatting standards, and follow themconsistently in all screens within a systemurelated screens in a system should always have standardtypes of information located in the same part of the screenve.g. screen titles, status messages, input fields, command inputlines, and error messages should consistently appear in thesame area of the screen throughout the systemuTipudesign one or more screen templates that follow rules ofscreen layoutufollow the templates[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 17Fonts and Stylesuminimize font changesutwo different fonts per screen is plentyufonts that look good on paper may not necessarily lookgood when displayed, and vice versaufor text to be viewed from a distance, avoid the use ofserifsvuse fonts like Helvetica or Arialuminimize style changesustyles are changes within a fontvitalics, bold, and

underliningvthey also include different character sizes-12 point, 14 point, 16 point, 18 point, 20 point, 24 point, 28 point[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 18CAPITALSuminimize the use of words in ALL UPPER-CASE(CAPITAL)utext in all uppercase is more difficult to read than text inmixed casesvstudies have shown that reading mixed text is about 13% fasterthan reading uppercase textuuse uppercase letters to designate titles, main categories,etc.uconsider the use of SmallCaps insteadvif available[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 19Captions and Fieldsudistinguish and segregate captions and fieldsua fill-in or display field should be easily distinguishablefrom its caption, label, or titleuconsider the use of contrasting features to make captionsand fields stand outvcolor, case, intensity, font, or reverse video[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 20

COURSE OFFERINGSQUARTER UNIVERSITYSPRING 96ALLSEQUENCE 02PROFESSOR ALLCOURSECOURSENUMBERNUMBERNAMECREDITSPROFESSORCOMP535DATASTRUC03 SMITHPSYC101INTRO03ADAMSBIOL201GENETICS03PAPERMANCOMP675HCI03MUSTILLOScreen Design Examples[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 21

Screen Design Examples[Mustillo]COURSE OFFERINGSQUARTER:SPRING UNIVERSITY:ALLSEQUENCE: 02PROFESSOR:ALLCOURSECOURSENUMBERPROFESSORNUMBERNAMEOF CREDITSCOMP 535 Data Structures 3J. SmithPSYCH 101Introduction to 3S. AdamsPsychologyBIOL 201Genetics 3A. PapermanCOMP 675Human-Computer 3P. Mustillo Interface DesignPress ARROW KEYS to scroll <RETURN> to go back ' 1999 Franz KurfessVisual Design Guidelines 22

Screen Design Examples[Mustillo]COURSE OFFERINGSQUARTER UNIVERSITYSPRING 96ALLSEQUENCE 02PROFESSOR ALLCOURSECOURSENUMBERNUMBERNAMECREDITSPROFESSORCOMP535DATASTRUC03 SMITHPSYC101INTRO03ADAMSBIOL201GENETICS03PAPERMANCOMP675HCI03MUSTILLO COURSE OFFERINGSQUARTER:SPRING UNIVERSITY:ALLSEQUENCE:02PROFESSOR:ALLCOURSECOURSENUMBERPROFESSORNUMBERNAMEOFCREDITSCOMP535 Data Structures 3J. SmithPSYCH101Introduction to 3S. AdamsPsychologyBIOL 201Genetics 3A. PapermanCOMP675Human-Computer 3P. Mustillo Interface Design

Press ARROW KEYS to scrollPress <RETURN> to go back ' 1999 Franz KurfessVisual Design Guidelines 23Text Layoutulabelsudescriptions of icons, functions, commands, ...umessagesuinformation from the system to the userupromptsuexpect a response from the useruinstructionsutell the user what to douusually longer than messages or promptsumay be interactiveuoften used in help or documentation features[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 24Labelsulabels should be self explanatoryuavoid acronyms on labelsuif you can, display the entire worduplace labels to the left in a tableuleft justify labels if you have more than oneudo not right align labelsuright-aligned labels produce a ragged marginvhard to scanuavoid placing labels on top of text fields or areas[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 25Example Labels[Mustillo]Phone Numbers

Home(514) 488-8848

Work(514) 765-7743Fa x(514) 761-8503 ' 1999 Franz KurfessVisual Design Guidelines 26Messagesudesirable properties of messagesubrief and conciseuaffirmative; avoid negative messagesuconstructive, not criticaluspecific and comprehensibleuimply that the user is in controluuse words consistent with a required actionu(a) poor: �landing gear is not down�(b) better: �landing gear is up�(c) best:�put the landing gear down�udesign the level of detail according to the user�sknowledge and experience[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 27Examples Messages[Mustillo]PoorBetter

The processing of the texteditor yielded 23 pages ofoutputOutput 23 pages

Error in SHIRT SIZE fieldError: SHIRT SIZE rangeis 14-18

Cannot exit before saving fileSave file before exiting

Bad/illegal/invalid field nameMaximum file name lengthis 8 characters

Syntax error 1542Unmatched leftparenthesis in line 210

' 1999 Franz KurfessVisual Design Guidelines 28Promptsuarrangement of promptsuwhen and where they will be neededuorder prompts chronologically or sequentiallyuformat prompts using blank spaces, justification, and visualcues for easy scanningudesign the level of detail according to the user�sknowledge and experience[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 29Prompts (cont.)uphrasing informationuuse the active tensevsay �Push the button�not �The button is to be pushed by the user�uphrase prompts as affirmative statementsvavoid negatives�Do x� vs. �Do not do y�uuse consistent terminologyvuse the word �type� when instructing a user to type somethingvuse �Press� for pressing a button or function key-user does not type the word �Return� when instructed to �Press Return�vreferences on labels should be exact-if a key is labeled �RTN�, don�t say, �Press Return�, say �Press RTN�-difficult to achieve since different users may have different keyboards[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 30Examples PromptsPoorBetter

returnPosition cursor,Press return to accept

SIZE:___SIZE:___ (14-18)

The message is sent bypressing TRANSMITTo send message,Press TRANSMIT

Do not return to menu beforecompleting entryComplete entry beforereturning to menu

Page forward after enteringaddressEnter address, thenpage forward

Press backtab to go up,tab to go downTo go up, Press BACKTABTo do down, Press TAB

For up, Press UFor down, Press D

' 1999 Franz KurfessVisual Design Guidelines 31Instructionsuarrangementukeep paragraphs shortu separate them by at least one empty lineulimit lines to 50-55 characters per linevor use double columns, 30-35 characters per columnuavoid right justification with unequal spacinguspace between lines should be equal to or slightly greaterthan character heightvapprox. double spaceduavoid line lengths of less than 26 charactersTO AVOIDTHAT THEYLOOK LIKECOLUMNS[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 32Instructions (cont.)uphrasing instructionsukeep text simple and clearuuse short sentences, and simple and familiar wordsuavoid hyphenationuavoid acronymsvunless you are sure that all users will understand them[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 33Layout of Numbersuright justify integersualign decimals in real numbersuavoid leading zeros when they are unnecessary andnon standardubreak up long numbers into groups of three or fourdigitsuuse standard separators when they applyuotherwise, use spaces[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 34Examples Numbers[Mustillo]PoorBetter

101001,00010,000101001,00010,000

100.0025.25635,432.481.45491 100.00 25.2563 5,432.48 1.45491

02/07/9610:01 p.m.2/7/96

002100013 2100 13

51484830388003648756 (514) 838-30381-800-364-8756

617395468602840555617-395-4686028-40-555

' 1999 Franz KurfessVisual Design Guidelines 35Coding Techniquesuuse attention-getting techniques appropriatelyuknow your screen device and what it is capable ofdisplayinguuse these techniques sparinglyublinking, reverse video, colorubold, italics, shadow,

underline, emboss, outline uCAPITALS and SmallCapsufont, size, shape, special characters and iconsuproximity, borders, depthusound, animation and motion[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 36Coding Techniques (cont.)ublinkinguuse to get the user�s attention. example: critical errormessagesurecommended blink rate of 2-5 Hz, with longer �on� periodthan �off� periodvminimum �on� period of 50%ureverse videoudark foreground on light background or light foreground ondark backgroundumay reduce legibility, so should not be overuseduvery powerful search cueumay be used to indicate that an item has been selected orto indicate an error field[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 37Coding Techniques (cont.)usizeucan be used as a cue for relative quantity or importanceulimit to 5 or less levels, because users have difficultydistinguishing more than fiveufontucan be used to draw attentionvuppercase characters, double-sized characters, italics, and otherdifferent fontsulimit to 2 different character types per screenusame limit applies to the use of different font sizes ' 1999 Franz KurfessVisual Design Guidelines 38Coding Techniques (cont.)ubolduuse to make things stand out, but avoid using more thanthree different levels of boldnessvusers don�t remember the differenceuunderliningucan be used to make an object or character stand out,especially key wordsumay reduce legibility, so it should be used sparingly ' 1999 Franz KurfessVisual Design Guidelines 39Coding Techniques (cont.)ushapeumay be used to impart different connotations, or implysome meaningvred octagon shape in north America is usually associated with stopsignsuspecial characters and iconsutags such as asterisks, arrows, and other graphicalsymbols, such as a small pointing hand can be used todraw attention to items on the screen ' 1999 Franz KurfessVisual Design Guidelines 40Coding Techniques (cont.)uproximityuproximity implies association, and leads the eye toconsider items as part of a wholeubordersuvery effective in identifying meaningful groups, creating asense of closure, and focusing attention ' 1999 Franz KurfessVisual Design Guidelines 41Coding Techniques (cont.)usounduvery effective in drawing attentionudifferent sounds can convey closeness, magnitude, weight,etc.ushould be unique when used in noisy environmentsuanimation and motionunot much known about the use of animation and motion asa screen design technique, although it would seem to be agood attention-getting or alert mechanismvthe human eye is very sensitive to motion in the peripheryve.g. insect on the wall ' 1999 Franz KurfessVisual Design Guidelines 42Coding Techniques (cont.)udepthudepth is a powerful cue in naturevmonocular depth cues include: (a), size, (b) interposition, (c)contrast, clarity, and brightness, (d) shadow, (e) textureumotion parallax is a another powerful depth cuevobserver moving relative to the environment in a train watching theoutside worldutechnology not yet capable of rendering true stereoscopicdepth on a display, although significant progress has beenmadevareas of wire-frame modeling, solid-object modeling, and volumetricrenderingvsome displays now support stereoscopic viewing

' 1999 Franz KurfessVisual Design Guidelines 43Coding Techniques (cont.)ucoloruvery powerful and salient coding technique that can beused in designing screensvto draw attentionvto communicate organizationvto indicate statusvto establish relationshipsvto express similarityvas a search cue ' 1999 Franz KurfessVisual Design Guidelines 44Color (cont.)umust be chosen and used with great carevthere are physiological as well as cultural differences in how weperceive colorsvwe perceive colors differently, depending on age, cognitivepreferences, exposure to color (adaptation)vcolor perception is subject to contextual effects, in which adjacentcolors influence one another. for example, a gray square on a darkbackground appears lighter and brighter than the same color on alight backgroundvcolor perception is subject to cognitive stereotypes-red -> danger-green -> go-yellow - warning, attention ' 1999 Franz KurfessVisual Design Guidelines 45Colors and Cultures[Mustillo]SOURCE: Courtney, A.J. (1986). Chinese population stereotypes: Color associations.Human Factors, 28(1), 97-100.ConceptColor%Color %SafeGreen62.2Green 61.4ColdWhite71.5Blue 96.1CautionYellow44.8Yellow 81.1GoGreen44.7Green 99.2OnGreen22.3Red 50.4HotRed31.1Red 94.5DangerRed64.7Red 89.8OffBlack53.5Blue 31.5StopRed48.5Red100.0 Chinese American

' 1999 Franz KurfessVisual Design Guidelines 46Colorvuse color sparinglyudesign first in monochromevoptimize other aspects of the screen layout designvadd color only where it adds valuevdon�t use color just because you have a color monitoruuse color to draw attention, to communicate organization,to indicate status, and to establish relationshipsuuse color to support search tasksvcolor is a more salient cue for search tasks and symbolidentification tasks than other cues, such as shape, size, orbrightnessvexample: counting the number of red objects vs. the number ofhorizontal bars, of small objects, or bright objectsuavoid using color in non-task-related waysuallow users to control color coding[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 47Color (cont.)udo not use color without some other redundant cueunot all users may have color displayeduredundant coding allows the interface to be ported to amonochrome workstation without losing importantinformationuapprox. 8% of males and 0.4% of females have some sortof color deficiencyuredundant coding enhances performance.ve.g. It is faster, easier, and less prone to error to search for a redsquare than to search for objects that are just red or just squares[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 48Color (cont.)uensure that colors differ in lightness and brightnessas well as in hueubrightness refers to the dimension of dull to brightuhue is the visual sensation that varies according towavelengthumost common types of color deficienciesvthe inability to discriminate red from green and blue from yellowuall sighted people can discriminate differences in lightnessvlightness refers to the range of achromatic colors ranging fromwhite through gray to blackuedges may not be clear for colors on a colored backgroundvcolors must contrast in brightness as well as hueubrightness is mediated by red and green photoreceptorsvcolors differing only in blue will not produce sharp edges[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 49Color (cont.)ube consistent with color associations that users havein their jobs and in their cultureuin Western cultures, the following associations are typicallyreinforced in traffic signals and consumer products.vGreen:Go, on, safevRed:Stop, hot, danger, on, emergencyvYellow:Caution, warning, warmvBlue:Cold, offuTip: Usevred for error messagesvyellow for warning messages[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 50Color (cont.)ulimit coding to 8 distinct colorsupreferably less than fourumake them as distinctive as possiblevspaced as far away as possible on the visible color spectrumuIf you use more than 6 or 8 colors, provide a legenduto assist users in learning and remembering color codesuavoid using saturated blues for text linesuvisual acuity is very poor for detailed, sharp, short-wavelength targetsuhe human eye is not good at focusing on bluevthe fovea (central area of the eye) is insensitive to blueuinsensitivity to blues increases with ageuthe lens becomes yellow with age[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 51Color (cont.)uchoose symbol and background colors carefullyu display color images on an achromatic background (black,gray, or white), or achromatic images on a colorbackgroundvimproves color contrast and color discriminationuensure greatest amount of contrast between foregroundand background under different lighting conditionsva light background could result in increased flicker and glaresensitivityuuse dark backgrounds for colored text, light backgroundsvespecially when presenting large graphical symbols[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 52Color (cont.)uconsider viewing angle and distance in assigningsymbol colorsuviewing angle and distance influence color perceptionuISO 9241 recommends the following:vavoid red and green beyond 40 deg. of viewing anglevavoid yellow beyond 50 deg. of viewing anglevavoid blue beyond 60 deg. of viewing anglevuse highly saturated colors and colors with high luminance contrastfor viewing distances beyond 60 cmvavoid saturated colors on dark backgrounds for viewing distancesbeyond 60 cm[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 53Color (cont.)ucolor sensitivity changes under different lightingconditionsureds and blues are more difficult to distinguish in dimlighting than greens and yellowsvthe eye is most sensitive to middle wavelengthsve.g. a red rose seen at dusk looks blackuolder users need higher brightness levels todistinguish colorsuthe overall sensitivity of the eye diminishes with ageuthe lens becomes less transparent[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 54Color (cont.)uavoid red and green in the periphery of largedisplaysuthe human eye is insensitive to red and green in theperipheryvespecially when these colors are highly saturated, and when thedisplayed targets or shapes are smallvsaturation refers to the number of different wavelengths present orcontributing to the sensation of colorufor the periphery of large displays, use yellow or blueuavoid the simultaneous display of highly saturated,spectrally extreme colorsve.g. red vs. blue, or yellow vs. purplevrequires constant re-focusing[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 55Color (cont.)uavoid the use of red and blue togetheruparticularly for textuthe eye cannot focus on both colors simultaneously.vthe use of both colors results in a perceptual phenomenon knownas chromostereopsisv i.e., red and blue text appears to lie in different depth planes[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 56Chromostereoscopic Effect[Mustillo]Red text appears to lie in one depth plane and blue text appears to lie in a different planeRed text appears to lie in one depth plane and blue text appears to lie in a different planeRed text appears to lie in one depth plane and blue text appears to lie in a different planeRed text appears to lie in one depth plane and blue text appears to lie in a different planeRed text appears to lie in one depth plane and blue text appears to lie in a different planeRed text appears to lie in one depth plane and blue text appears to lie in a different plane ' 1999 Franz KurfessVisual Design Guidelines 57Summary Screen Layout and DesignThe format and content of informationdisplayed on the screen is veryimportant in determining the success ofa user‘s interaction with a system.Three aspects to consider in screen design:¥Amount of Information Presented (Density)¥Presentation of Information (Spatial Relationships)¥Complexity of Information Presented

' 1999 Franz KurfessVisual Design Guidelines 58Summary Screen Layout and Designudensityuminimize the amount of information by presenting onlywhat is absolutely necessary to the uservconcise wording, familiar data formatsvavoidance of unnecessary detailuspatial relationshipsugrouping similar items in a display for readabilityvcan highlight relationships between different groups of data-color coding, graphic borders, highlighting, reverse video, brightnessdifferencesucomplexity of information presentedukeep the layout of your screens as simple as possibleuapplies equally for all types of screen-based interfacesvincluding Web pages[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 59Implications for Windows Designuorganizational and visual/graphics principles andguidelines for the design of screen-based UIs applyequally to the design of window systemsuguidelines are available on various aspects ofwindows design[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 60Buttonsuuse sufficiently large buttonsuas the size of a button increases, so does the speedneeded to traverse to and select that buttonuas the distance of the mouse movement decreases, speedalso increasesvknown as Fitt�s Lawvkeep buttons large and mouse movement distance shortukeep button size consistentumake all of the buttons in a dialog box the same sizevunless you have a very good reason for different sizes

' 1999 Franz KurfessVisual Design Guidelines 61Buttons (cont.)ugroup buttons togetherusimilar functionsuto leave the window (OK, Cancel)udestructive actions (Delete)uuse heading style capitalization for button labelsucapitalize the first letter in each word of the button labelulabel and order buttons consistentlyu clear and concise button labelsuaffirmative buttons to leave a window (OK)ucanceling actions to leave the window (Cancel)uassign a nondestructive default buttonugray out or (dim) a button if it is unavailable[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 62Menusumaintain a consistent relationship with defined termsand conceptsuuse ellipsis (�) in items to indicate additional userinputuOpen �, Save as �, Print �udisable inactive menu itemsue.g., by dimming them outuuse check boxes to enable and disable attributesuuse radio buttons to select one item amongattributes or commands[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 63Menus (cont.)udo not use more than two levels of cascading menusuthey add complexity to the interfaceuthey hide choices by adding a layer to menusuare physically more difficult to useushould only be used to minimize screen real estateuorganize cascading menus with a logical relationshipto the menu items they containuminimize the use of pop-up windows[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 64Implications for Web Designunavigationudesign elementsuuser interface consistency[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 65Importance of the Webuthe Web is developing at an astonishing paceuone new Web user every 2 seconds (Forrester Research)umore than 150 million Web users worldwide todayuguidelines for Web interfaces are beginning toemergeue.g., Apple, IBM, Microsoft, Sun, � (see below)uthe Web is about:ucontentvhigh-quality content is a major driver for attracting repeat visitorsureach (extensibility)unavigationuweb design is an evolutionary step in thedevelopment of UI design[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 66NavigationuWeb sites or browsers should provide the followingnavigational aidsuactive table of contentsubacktracking and history listsusite mapsubookmarks (Netscape) or favorites (Internet Explorer)utimestampsvdon�t date information unless it is updated frequently.vusers will immediately leave a site if they see that the information isnot currentvtell users how often information will be updatedvupdate information automatically[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 67Design Elementsuframesusplitting a page into frames can be confusing for usersuframes may break the fundamental model of the Web pagevprintouts become more complicatedvpredictability of user actions may be violatedugraphicsunot all Web designers are good graphic designersuusers have a short attention spanvif downloading a page takes longer than 10 sec., they may leaveuscrolling text, marquees, animated sequencesudo not include page elements that move all of the time voverwhelming to many users, annoying to othersvusers are very sensitive to motion in the periphery-may distract from real task[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 68Design Elements (cont.)ucomplex URLsuusers try to infer the structure of Web sites from the URLushould contain a human-readable directory and file namesvreflect the nature of the information spacevevery page should have a link to the home page and an indication of where it will fit within the structure of the informationspaceuorphan pagesumake sure that all pages include a clear indication of whatWeb site they belong tovusers may access pages without going through the home pageuoutdated informationuinformation on the Web can get out of date very quickly[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 69Navigationunavigational supportuusers don�t know as much about your site as you douusers get lost very easilyvprovide a site mapv let your users know where they are and where they can gouprovide a good search mechanismvsometimes even the best navigation mechanism is not enoughvsome users want to jump directly to the page they needunon-standard link colorsulinks to pages that users have not seen are typically blueulinks to previously visited pages are typically purple(magenta) or redvbe consistent; don�t mess around with color standards[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 70Textuusers don�t read Web pages; they scan themureading speeds are >25% slower from computer screenuuse scannable textvhighlighted words and descriptive linksvmeaningful sub-headingsvbulleted listsvone idea per paragraphvinverted pyramid style - present conclusion firstvcut down on the number of wordsuusers don�t like to scroll.uonly 10% of users scroll beyond the information that isvisible on the screen when a page comes upuall critical information should be on the top part of the page[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 71Usersutry to determine who your users are, and what theywant to dounew or returning visitorubrowsing, seeking support, want to purchase, download,contact?u label all linked graphicsu do not leave orphan links, where users must back upvalways provide at least a way to get back to the home pageukeep links updatedvcheck to see if all links are still validuregularly check your links in your documentsvmake sure they go to the right placeustandard text link colors and highlighting[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 72Navigation Supportuhelp users navigate and searchuclear and intuitive site infovtable of contents, index, site map, �What�s New?�uunderstandable and obvious visual navigation schemeve.g. buttons, text, site mapsusearch for topics, keywords, names, etc. within a siteutopics keywords, and names into Web search enginesuprovide navigational aids on every pagevmake it possible for users to jump right to any page from a Websearch engineu let users know where a navigation link will take them[Mustillo]

' 1999 Franz KurfessVisual Design Guidelines 73Tasksuhelp users perform tasks quicklyureduce search time, time to load pages and images,scrolling within a pageulimit number of choices at a levelugroup important choices togetheruorganize information and pages to anticipate frequenttasksuprovide an overview of the siteudon�t use bleeding-edge technology without a good reasonve.g. use VRML only if you have information that maps naturally to3D space[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 74Contentukeep content simplevaddress only one topic per pageuget to the content quicklyvplace important content at the top of Web hierarchy and at the topof the Web pageuprovide printable versions of Web pages[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 75Consistencyuestablish user interface consistencyucorporate identityvgraphics, images, colors, fonts, and layoutvuse the corporate identity consistently throughout the Web interfaceuconsistent link and navigation schemes throughout theWeb interfaceuconsistent headings and terminologyudisplay background infovURL, copyrights, e-mail addresses, etc.udesign the interface to be easily accessible and usablevdifferent browsers, platforms, screen resolutions[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 76Web ResourcesuApple Web Design Guidevhttp://www.cybertech.apple.com/hi/web_design/web_design.htmluIBM Human-Computer Interactionvhttp://www.ibm.com/ibm/hci/designer/uides/webdesign.htmluSun Microsystemsvhttp://www.sun.com/styleguideuYale Center for Advanced Instructional Mediavhttp://info.med.yale.edu/caim/StyleManual_Top.HTMLuJakob Nielsen, Alertboxvhttp://www.useit.com/alertboxuSusan Weinshenkvhttp://www.weinshenk.com[Mustillo] ' 1999 Franz KurfessVisual Design Guidelines 79Important Concepts and Termsuusabilityuuser interface designu

What

You

SeeIsWhat

Get�(WYSIWYG)uwindowuergonomicsugraphical user interface (GUI)uhuman factors engineeringuinput/output devicesuknowledge management ' 1999 Franz KurfessVisual Design Guidelines 80Chapter Summaryuthe visual channel has the highest informationtransfer capacity from computer to user, and is usedvery extensivelyuvisual aspects have a strong influence the usabilityand acceptance of user interfacesuthe arrangement of design elements, text, numbers,and graphics determine the appearance of the visualdisplayucolor and coding techniques can be used to guidethe attention of the user

' 1999 Franz KurfessVisual Design Guidelines 81